Author's response to reviews

Title: Association between single nucleotide polymorphisms in the mu opioid receptor gene (OPRM1) and self-reported responses to alcohol in American Indians

Authors:

Cindy L Ehlers (cindye@scripps.edu)
Penelope A Lind (penelopL@qimr.edu.au)
Kirk C Wilhelmsen (kirk@med.unc.edu)

Version: 5 Date: 17 April 2008

Author's response to reviews:

We have corrected most of the formatting changes you required. Authors have been listed exactly as they are in the submission system; each affiliation has been written on a new line; the running title has been removed; website addresses have been removed from the text and cited in the main reference list only. In the preparation of the manuscript, a referencing software was not used. You indicate that there is a discrepancy between the number of references in the manuscript list and that extracted by your system. There are additional references in Table 1, which may account for the discrepancy. Figure has been cropped to minimize white space around image as much as feasible.